Author's response to reviews

Title: The prevention of contrast induced nephropathy by sarpogrelate in patients with chronic kidney disease: a study protocol for a prospective randomized controlled clinical trial

Authors:

Kyungil Park (cardiopark@gmail.com)
Woo-Young Chung (wychung@paran.com)
Jae-Bin Seo (yongseongs@paran.com)
Sang-Hyun Kim (shkimmd@snu.ac.kr)
Joo-Hee Zo (jooheezo@hanmail.net)
Myung-A Kim (kma@brm.co.kr)
Young-Bae Park (parkyb@snu.ac.kr)

Version: 7 Date: 5 December 2010

Author's response to reviews: see over
Dec 4th, 2010

Editor, *Trials*

Dear editor:

We would like to thank you and the reviewers of “*Trials*” for taking the time to review our article. We have made a correction and clarification in the manuscript after going over the reviewer’s comments. The change is summarized below.

**Authors’ response to the Reviewer’s comment**

**Reviewer:**

**Specific comments to the author:**

**Major revisions (We require the author to make these changes)**

*Figure 1:* Figure 1 has not been cited in the text. Each figure should be cited at least once in the text. Please add this citation.

We have added the sentence that cited Figure 1 as the reviewer suggested (Page 6, line 2 and 3).

Please take this opportunity to check your manuscript for any typographical errors.
and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Typographical errors and wrong words have been corrected as the reviewer has suggested (Page 3, line 11 and 14; Page 5, line 5, 13 and 14; Page 6, line 10, 21 and 22; Page 7, line 23; Page 8, line 8 and 13; Page 10, line 3, 14 and 15; Page 12, line 17 and 22; Page 13, line 4, 8, 12, 16, 20, 24; Page 14, line 2, 7, 12, 15, 18, 21, 24; Page 15, line 5, 8, 12, 15, 19, 23; Page 16, line 1, 4, 7, 13, 18, 21; Page 17, line 1 and 4).

We hope the revised manuscript will better meet the requirements of your journal for publication. We thank the editor and reviewer of “TRIALS” once again for the constructive review of our paper.

Sincerely yours,

Woo-Young Chung MD
Associate Professor
Division of Cardiology, Department of Internal Medicine
Seoul National University College of Medicine
Seoul National University Boramae Medical Center

Shindaebang 2-Dong 425, Boramae Gil 39, Dongjak-Gu, Seoul, 156-707, Korea

Tel : 82-2-870-2218

Fax : 82-2-870-3866

E-mail: wychung@paran.com